Author’s response to reviews

Title: Heart rate variability after radiofrequency ablation of epicardial ganglionated plexuses on the ovine left atrium

Authors:

Vilius Kviesulaitis (viliuskviesulaitis@yahoo.com)
Aras Puodziukynas (aras.puodziukynas@gmail.com)
Vytautas Zabiela (vytautas.zabiela@kaunoklinikos.lt)
Tomas Kazakevicius (kazakev@gmail.com)
Raimundas Vaitkevicius (raimvaitk@gmail.com)
Evaldas Dirzinauskas (info@eveterinaras.lt)
Vytenis Semaska (vytenis_semaska@yahoo.com)
Antanas Strazdas (medstrazdas@gmail.com)
Ruta Unikaitė (runikaite@gmail.com)
Dainius Pauza (dainius.pauza@lsmuni.lt)

Version: 4 Date: 29 Nov 2017

Author’s response to reviews:

Hello, I apologize for this many corrections.

Section 'Ethics approval' is amended with full and CORRECT name of ethic committee. Apologize again for a wrong translation. Original translation was slightly different compared to what they give on their site (http://vmvt.lt/?language=en) but sadly I did not bother myself to check for it in the first time. Manuscript text was corrected accordingly.

Now it goes as follows:<<<Ethics approval: study permission No.:33JV-62 was issued by the State Food and Veterinary Service after the approval of Lithuanian Ethics Committee on the use of laboratory animals under the State Food and Veterinary service.>>>